Author’s response to reviews

Title: Effectiveness of a Community Program for Older Adults with Type 2 Diabetes and Multimorbidity: A Pragmatic Randomized Controlled Trial

Authors:
John J. Miklavcic (miklavcic@chapman.edu)
Kimberly D. Fraser (kdfraser@ualberta.ca)
Jenny Ploeg (ploegj@mcmaster.ca)
Maureen Markle-Reid (mreid@mcmaster.ca)
Kathryn Fisher (fisheka@mcmaster.ca)
Amiram Gafni (gafni@mcmaster.ca)
Lauren E. Griffith (griffith@mcmaster.ca)
Sandra Hirst (shirst@ucalgary.ca)
Cheryl A. Sadowski (cherylas@ualberta.ca)
Lehana Thabane (thabanl@mcmaster.ca)
Jean A. C. Triscott (jean.triscott@ualberta.ca)
Ross Upshur (ross.upshur@utoronto.ca)

Version: 1 Date: 27 Sep 2019

Author’s response to reviews:

Many thanks for your note re necessary corrections.
We have ensured the manuscript adheres to CONSORT guidelines.
We have attached a CONSORT checklist.
We have added the date of trial registration.
We have renamed our ethics section to ethics approval and consent to participate
We have renamed the section Availability of data to Availability of data and materials.